Author’s response to reviews

Title: Do we understand the intervention? What complex intervention research can teach us for the evaluation of Clinical Ethics Support Services (CESS).

Authors:

Jan Schildmann (jan.schildmann@medizin.uni-halle.de)
Stephan Nadolny (stephan.nadolny@fh-bielefeld.de)
Joschka Haltaufderheide (joschka.haltaufderheide@rub.de)
Marjolein Gysels (mhgysels@gmail.com)
Jochen Vollmann (jochen.vollmann@rub.de)
Claudia Bausewein (Claudia.Bausewein@med.uni-muenchen.de)

Version: 3 Date: 11 Jun 2019

Author’s response to reviews:

Dear Editorial Team,

thank you for your comments.

Please find below your comment followed by the authors' response.

Editor Comments:

--Unfortunately the BMC Series does not accept footnotes for publication. Please remove the footnotes and include the information in a different format. Please note BMC Series does publish endnotes.

ANSWER: We have removed the footnotes and where necessary integrated the information in the main text.
-- Please note that footnote 2 on the first page still reads 'XXX', if you do move this information please complete this sentence when moving it.

ANSWER: The text of footnote 2 has been completed and integrated in the main text.

-- In order to be in line with journal requirements, please include the following headings within your mainscript: Background, Main Text (which can be broken into sub-sections), Conclusions

ANSWER: The heading have been changed in accordance with the journal's requirements.

-- Thank you for providing a response to the reviewers. As this document is no longer required at this stage of the publication process, please remove it from your submission’s supplementary files.

ANSWER: We have deleted the document.

-- At this stage, please upload your proofread manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional-supplementary files are cited within the text.

ANSWER: The manuscript has been uploaded as clean version. All relevant other files have been included.